Author’s response to reviews

**Title:** QUERI Series: EQUIP: Implementing chronic illness care principles and applying formative evaluation to improve care for schizophrenia

**Authors:**

- Alison H Brown (alisonh@ucla.edu)
- Amy N Cohen (ancohen@ucla.edu)
- Matthew J Chinman (chinman@rand.org)
- Christopher Kessler (Christopher.Kessler@va.org)
- Alexander S Young (ayoung@ucla.edu)

**Version:** 7  **Date:** 7 February 2008

**Author’s response to reviews:**

The title and abstract are now identical in the manuscript and in the submission system.

Blank pages have been removed.

The website address from Table 2 has been removed from the table and cited from the main reference list instead.

The website address from p.14 is now cited from the main reference list only. References were entered using the software and long website addresses were avoided.